Author's response to reviews

Title: Evaluation of Microscopic Observation Drug Susceptibility assay for diagnosis of multidrug-resistant tuberculosis in Viet Nam

Authors:

Dang TM Ha (hadtm@oucru.org)
Nguyen TN Lan (ngoclan0456@yahoo.com)
Marcel Wolbers (mwolbers@oucru.org)
Vo S Kiet (kietvs@oucru.org)
Hoang TT Hang (hanghtt@oucru.org)
Nguyen H Duc (drnguyenhongduc@yahoo.com)
To M Huong (Tomyhuong@gmail.com)
Vuong TM Bach (bsminhbach@yahoo.com)
Nguyen TP Thao (thaongth@yahoo.com)
Tran V Quyet (quyettran80@gmail.com)
Nguyen TB Tuyen (tuyen135200@yahoo.com)
Vo T Ha (kathy201004@gmail.com)
Nguyen T Nho (nhont@oucru.org)
Dai V Hoa (viethoa.pnt@gmail.com)
Phan TH Anh (phanthihoanganh@gmail.com)
Nguyen H Dung (bshuydung@yahoo.com)
Jeremy J Farrar (jfarrar@oucru.org)
Maxine Caws (mcaws@hotmail.com)

Version: 2 Date: 28 July 2011

Author's response to reviews:

The following requested revisions have been made in the manuscript:

1. Author's contribution has been added
2. Competing interests statement has been added.
3. The funding statement has been clarified.
4. Separate emails of authors have been added